Author’s response to reviews

Title: Stress on facial skin of Class III subjects during maxillary protraction: a Finite Element Analysis

Authors:

Francesca Gazzani (francescagazzani@hotmail.it)
Chiara Pavoni (dott.chiarapavoni@gmail.com)
Paola Cozza (profpaolacozza@gmail.com)
Roberta Lione (robertalione@yahoo.it)

Version: 3 Date: 31 Jan 2019

Author’s response to reviews:

Editor Comments:

dear authors you have successfully modified your paper according to all our requests and the paper can be accepted for publication now, but there is one little (last) modification needed, before the paper can be published.

in the discussion section, the limitations should not stand alone in a separate section, but in a separate paragraph INSIDE the discussion. the discussion cannot be structured. no subtitles nor subsections here.

so please erase the subtitle "Limitations" because discussion is one and cannot be separated nor structured, i jest asked you to add a paragraph about limitations and you did it, no need to design subsections

inside the discussion. please erase line 162 and resubmit your paper. thank you.

We are sorry for the inconvenience. Thank you for the comment. As suggested the subtitle "limitation" was delated.